Author's response to reviews

Title: Cervical dysplasia and cancer: use of hormonal contraceptives in Jamaican women

Authors:

Norma McFarlane-Anderson (norma.mcfarlaneanderson@uwimona.edu.jm)
Patience E Bazuaye (patebazu@yahoo.com)
Maria D Jackson (maria.jackson@uwimona.edu.jm)
Monica Smikle (monica.smikle@uwimona.edu.jm)
Horace M Fletcher (horace.fletcher@uwimona.edu.jm)

Version: 5  Date: 29 May 2008

Author's response to reviews:

Biomed Central Editorial Production Team

Please accept my apologies.

I have changed my registration with BMC to read Norma McFarlane-Anderson. I hope this is now acceptable.

On the cover page I have added my name as the corresponding author.

Page now reads: Cervical dysplasia and cancer and the use of hormonal contraceptives in Jamaican women.

Norma McFarlane-Anderson§, Patience E Bazuaye1, Maria D Jackson2, Monica Smikle3 Horace M Fletcher4

1 Department of Basic Medical Sciences, University of the West Indies, Mona Campus, Kingston, Jamaica
2 Department of Community Health & Psychiatry, University of the West Indies, Mona Campus, Kingston, Jamaica
3 Department of Microbiology, University of the West Indies, Mona Campus, Kingston, Jamaica
4 Department of Obstetrics & Gynaecology, University of the West Indies, Mona Campus, Kingston, Jamaica

§ Corresponding author

Professor Norma McFarlane-Anderson

Biochemistry Section